Author’s response to reviews

Title: Living with osteoarthritis is a balancing act: An exploration of patients' beliefs about knee pain

Authors:
Ben Darlow (ben.darlow@otago.ac.nz)
Melanie Brown (melanie.brown@otago.ac.nz)
Bronwyn Thompson (bronwyn.thompson@otago.ac.nz)
Ben Hudson (ben.hudson@otago.ac.nz)
Rebecca Grainger (rebecca.grainger@otago.ac.nz)
Eileen McKinlay (eileen.mckinlay@otago.ac.nz)
J Haxby Abbott (haxby.abbott@otago.ac.nz)

Version: 2 Date: 16 Apr 2018

Author’s response to reviews:

Dear Dr Leung

Thank you for providing this additional suggestion.

We have numbered the two themes as requested by the reviewer. We are happy to include numbers if these are acceptable to the journal, but also consider that highlighting the theme titles in italics (as is the case in the formatted version of the manuscript) is sufficient should the addition of numbers result in the abstract exceeding the word count.

Best regards

Ben Darlow